Author's response to reviews

Title: Do gender differences in primary PCI mortality represent a different adherence to guideline recommended therapy? A multicenter observation.

Authors:

Ralf Birkemeyer (rbirkemeyer@t-online.de)
Henrik Schneider (henrik.schneider@med.uni-rostock.de)
Andreas Rillig (arillig5@yahoo.de)
Juliane Ebeling (jule.ebeling@web.de)
Ibrahim Akin (ibrahim.akin@med.uni-rostock.de)
Stefan Kische (stefan.kische@me.uni-rostock.de)
Liliya Paranskaya (liliya.paranskaya@me.uni-rostock.de)
Werner Jung (werner.jung@sbk-vs.de)
Hueseycin Ince (hueseycin.ince@me.uni-rostock.de)
Christoph A Nienaber (christoph.nienaber@med.uni-rostock.de)

Version: 2 Date: 28 March 2014

Author's response to reviews:

1) Email addresses of all co-authors were included on the title page.
2) The name of the ethics committee was included
3) Page and line numbers were included in the manuscript
4) The figure legend was taken out of the manuscript and will be uploaded as a separate document.